

Received: 05 September 2017 Accepted: 27 November 2017 First Published: 06 December 2017

*Corresponding author: Dag Øivind Madsen, School of Business, University College of Southeast Norway, Bredalsveien 14, 3511 Hønefoss, Norway E-mail: dagm@usn.no

Reviewing editor: Bisogno Marco, University of Salerno, Italy

Additional information is available at the end of the article

MANAGEMENT | RESEARCH ARTICLE The diffusion of Lean in the Norwegian municipality sector: An exploratory survey

🔆 cogent

business & management

Dag Øivind Madsen^{1*}, Stig Risvik¹ and Tonny Stenheim²

Abstract: The use and application of Lean in the public sector has attracted considerable interest in recent years. In Norway, the concept of Lean has lately been hugely popular, not only among private sector firms, such as manufacturers, but increasingly also in the public sector where Lean has been proposed as a management tool that can improve the productivity and performance of, for example, municipalities. There have been several initiatives suggesting and recommending Lean adoption and implementation in the municipality sector. However, little is known about how and to what extent Lean is currently used by Norwegian municipalities. Therefore, this paper examines the diffusion of Lean among Norway's 426 municipalities, by means of an exploratory survey that yielded a response rate of about 25 percent. The survey results generally show that the awareness and adoption of Lean has increased quite sharply in recent years. Most municipalities have recently adopted and started using the Lean concept, and the concept appears to be in a relatively early stage of its lifecycle in the Norwegian municipality sector. When it comes to implementation of the Lean concept, the municipalities vary in terms of how they apply and utilize Lean. For example, the survey data indicate that municipalities use a broad spectrum of Lean-related tools and techniques. Finally, while the perceived effects of Lean are generally positive, there are also a range of issues and challenges associated with the implementation of Lean in municipalities.

ABOUT THE AUTHORS

Dag Øivind Madsen holds a PhD from the Norwegian School of Economics and is currently a faculty member in the Department of Strategy and Finance at the University College of Southeast Norway.

Stig Risvik holds a MSc in Business Administration from the University College of Southeast Norway.

Tonny Stenheim holds a PhD from the Copenhagen Business School and is currently a faculty member in the Department of Accounting, Auditing and Business Analytics at BI Norwegian Business School.

PUBLIC INTEREST STATEMENT

Lean is a management concept which is inspired by ideas and practices developed by the Toyota car company several decades ago. The overall aim of Lean is to reduce waste and improve the productivity of organizations. Since its inception around 1990, Lean has received much attention by academics and practitioners alike. In recent years, Lean has increasingly been applied in public sector organizations, for example, municipalities. This study examines the current diffusion and application of Lean in the Norwegian municipality sector, by means of an electronic survey. The results generally show that the awareness of and adoption of Lean has increased considerably in recent years. Overall, the respondents are satisfied with the use of Lean, and foresee continued use of the concept. At the same time, users of Lean also report implementation challenges related to, for example, organizational culture and managerial engagement.





 \otimes 2017 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

Subjects: Operations Management; Public Management; Organizational Studies; Organizational Change

Keywords: Lean; municipalities; management concept; adoption; diffusion; implementation; effects; experiences

1. Introduction

1.1. The concept of Lean in the public sector

The concept of Lean was introduced around 1990 (Krafcik, 1988; Womack, Jones, & Roos, 1990). Lean draws on insights and practices developed by the Toyota auto company decades earlier. Several commentators have noted that the Lean concept is vague and open to interpretation, and that there is little agreement among academics and practitioners about the exact meaning and content of the concept (Benders & van Bijsterveld, 2000; Brännmark et al., 2012; Wittrock, 2015). As pointed out by Ingvaldsen, Ringen, and Rolfsen (2014), Lean can be understood in four different ways; as an organizational trend, a management philosophy, a set of principles, or a set of practices. For example, the latter understanding of Lean highlights the concept's direct manifestations in organizations, as Lean users often use and apply different tools from the "Lean toolbox" (e.g. Poka-Yoke, Kanban, SMED, and Value Stream Mapping).

Since its introduction, Lean has attracted attention all over the world. As Wittrock (2015) points out, Lean has been positioned as a "world changing" management concept. Proponents of Lean (e.g. Lean "gurus" and consultants) generally argue that the concept can help organizations improve or-ganizational performance and productivity by reducing waste (e.g. time and resources spent on unnecessary and non-value-generating activities and processes). However, it has not been all "guruhype" and media attention. Many studies also show that Lean has had a considerable impact on organizational practice (Benders, 1996; Benders & van Bijsterveld, 2000; Ingvaldsen & Benders, 2016; McCann, Hassard, Granter, & Hyde, 2015; Wittrock, 2015).

In the beginning, Lean was primarily developed with the aim to improve the performance of private business organizations. Over time, the use of Lean has also spread like wildfire to the public sector. As pointed out by Bergström and Lapsley (2017, p. 17), the increased use of Lean in public sector management can be viewed as "a classic case of the mimicry of private sector practice."

In recent years, there has been much debate about how and to what extent the Lean concept can be successfully adapted and applied in public sector organizations, for example, municipalities (Bateman, Radnor, & Glennon, 2018; Bhatia & Drew, 2006; Drotz, 2014; Holmemo & Ingvaldsen, 2018; Holmemo, Ingvaldsen, & Benders, 2017; Martin, 2018; Radnor & Boaden, 2008; Radnor & Osborne, 2013; Radnor, Walley, Stephens, & Bucci, 2006; Scorsone, 2008; Suarez Barraza, Smith, & Mi Dahlgaard-Park, 2009).

While proponents of Lean are optimistic, some commentators, such as Bergström and Lapsley (2017, p. 17) are seemingly not fully convinced about the merits of Lean in the public sector, and suggest that Lean could be perhaps the latest in a long line of public sector management fads: "Indeed, there is almost an entire industry of management consultants, authors of guidebooks and a support industry of experts selling the merits of this latest fad in public services management which is, in fact, presented as novel and new despite being a recycled idea from decades earlier."

1.2. Purpose and contribution

Several recent literature reviews have shown that there is an emerging but small body of research on the diffusion and use of Lean in municipalities (Brännmark, 2011, 2012; Drotz, 2014). Bergström and Lapsley (2017, p. 19) suggest that the concept and practice of Lean in Scandinavian public services is under-researched, in particular issues related to the effectiveness and efficacy of the adoption of Lean. Therefore, the overall aim of this paper is to make a contribution to the literature on the diffusion and use of Lean in municipalities. The paper is guided by the following overall research question: To what extent and in what ways is the Lean concept diffused and used in the Norwegian municipality sector? The paper attempts to answer this question by carrying out an empirical study drawing on survey data from the Norwegian municipality sector. By doing so, the paper also contributes specifically to our knowledge about the use of the Lean concept in the Norwegian context.

Obtaining an updated picture of the impact of Lean in Norway is in and of itself interesting, since the Lean concept has attracted much attention and interest in Norway in recent years, not only in the private sector, but also in the public sector (Andersen & Røvik, 2015; Aspøy, 2014; Ingvaldsen, Rolfsen, & Finsrud, 2012; Rolfsen, 2014). For example, a recent survey of Lean in Norwegian organizations has shown that Lean is widely used among manufacturing, construction, as well as accounting and financial services firms (Madsen, Storsveen, Klethagen, & Stenheim, 2016).

However, our knowledge about the extent to which Lean is used in the Norwegian municipality sector, is quite limited. Two surveys of Lean, conducted more than five years ago, show that Lean started getting traction in Norwegian municipalities in the early 2010s (Dolva, 2011; Schie, 2012). There is reason to believe that a new survey could provide an updated picture and be able to cast light on developments in terms of Lean adoption, diffusion, and implementation in Norway.

1.3. Structure

The remainder of the paper is organized as follows. Section 2 provides a brief review of previous research on the Lean concept in the public sector, with an emphasis on the municipality context. Section 3 describes the methods and data used. Section 4 provides the main results, focusing on adoption, implementation, and effects of Lean. Section 5 discusses the main findings in relation to the extant literature on Lean in the municipality sector. The last section is dedicated to conclusions, as well as discussions of limitations and areas for future research.

2. The Lean concept in the public sector

This section provides a brief review of extant related research on Lean in the public sector, starting with a description of the importance of the concept for the public sector, before continuing with a deeper description of the case of municipalities.

2.1. Lean in the public sector

The early books and articles about Lean published during the 1990s focused primarily on the concept's application in the private sector, for example among manufacturing or construction firms (Alarcón, 1997; Liker, 1997; Womack & Jones, 1996; Womack et al., 1990). However, as mentioned in the introduction, Lean has over time, and particularly during the 2000s, been put forth as a viable management concept with a high degree of applicability in the public sector.

The demand for new innovative management concepts has been fueled by the many New Public Management (NPM) inspired reforms and modernization efforts (Hood, 1995; Hyndman & Lapsley, 2016) that have swept public sector organizations in recent years. For example, in the Scandinavian countries there has been a strong focus on modernizing the public sector to reduce waste and to improve productivity and performance (Bergström & Lapsley, 2017; Lapsley & Knutsson, 2016). Lean has emerged as one of the management concepts currently used as a part of NPM-inspired reforms and modernization efforts. For example, Bergström and Lapsley (2017, p. 17) argue that the fundamental aim of Lean, "to provide more with less resources has become something of a mantra for public sector modernisers."

The usefulness and applicability of Lean in the public sector is currently subject of much debate in the academic literature (Bateman et al., 2018; Holmemo & Ingvaldsen, 2018; Holmemo et al., 2017; Martin, 2018; McCann et al., 2015; Neumann, Mothersell, & Motwani, 2014; Radnor & Boaden, 2008; Radnor et al., 2006; Rahbek Gjerdrum Pedersen & Huniche, 2011a). For example, some researchers

have dealt with the issue of how the Lean concept has to be adapted when it is applied in a public sector context (Bhatia & Drew, 2006; Radnor & Walley, 2008). This is an important issue since research has shown that the Lean concept can be defined, interpreted and translated in different ways (Benders & van Bijsterveld, 2000; Brännmark et al., 2012; Morris & Lancaster, 2006; Pettersen, 2009; Wittrock, 2017, 2017). In a public sector context, researchers have noted the fluidity and negotiated nature of the Lean concept (Rahbek Gjerdrum Pedersen & Huniche, 2011b).

2.2. Lean in municipalities

Several recent literature reviews have shown that there is an emerging but still small body of research on the use of Lean in municipalities (Brännmark, 2011, 2012; Drotz, 2014). While there are some studies that have been carried out in other European countries, such as England (Radnor & Walley, 2008) and Spain (Suarez Barraza et al., 2009), the rest of this section will focus narrowly on research that has been carried out in a Scandinavian context (i.e. Sweden, Denmark, and Norway).

Studies which have been carried out in these countries are deemed to be of particular relevance since these countries can be characterized as Social Democratic welfare states where the public sector plays a big and active role in people's lives (Esping-Andersen, 1990). While previous research has shown that these three countries have had different responses to New Public Management-inspired reforms (Green-pedersen, 2002), it can be argued that these countries still share many commonalities in terms of size, politics, culture, and management style. Therefore, a review of past work carried out in a Scandinavian context helps contextualize the current study's overall aims and research question.

Previous research suggests that the Lean concept has taken different trajectories in the municipality sectors of the three Scandinavian countries (Arlbjørn & Freytag, 2009; Brännmark, 2011, 2012; Ringen & Rolfsen, 2014; Stentoft Arlbjørn, Vagn Freytag, & de Haas, 2011). It has been noted that Lean got a relatively early start in the Danish and Swedish municipality sectors (Ringen & Rolfsen, 2014).

For example, the Lean concept has received much attention in Denmark (Rahbek Gjerdrum Pedersen & Huniche, 2011a, 2011b), and has been used in Danish municipalities for some time (Ringen & Rolfsen, 2014, p. 121). Danish municipalities were inspired by Danish private sector organizations' experimentation with Lean (Ringen & Rolfsen, 2014). In a survey, Arlbjørn and Freytag (2009, p. 64) found that roughly 70 percent of Danish municipalities were using Lean in some way or another. Around this time, Lean was viewed favorably by Danish politicians and local government as part of a strategy to reduce costs and improve efficiency (Stentoft Arlbjørn et al., 2011), and as "an important cornerstone for increasing productivity" (Stentoft Arlbjørn et al., 2011, p. 286).

In Sweden, Lean has also received much attention and interest (Larsson, 2012; Modig & Åhlström, 2014), including within the public sector where the concept has "spread like wildfire" (Hallström & Thedvall, 2015; Thedvall, 2015, 2017). A couple of recent studies have shown that Lean is used within different parts of the Swedish public sector (Drotz, 2014; Innovationsrådet, 2012), and the concept has been quite extensively adopted and implemented in the municipality sector (Brännmark, 2011, 2012).

Compared with their Danish and Swedish counterparts, Norwegian municipalities have been relatively late movers in terms of adopting and implementing Lean. However, in recent years Lean has attracted much attention in the Norwegian public sector (Aspøy, 2014; Ringen & Rolfsen, 2014). For example, just in the last few years, the Norwegian government has published two Official Norwegian Reports¹ highlighting the need to improve productivity and efficiency in the public sector (NOU 2015:1, 2015; NOU 2016:3, 2016).

The Norwegian Association of Local and Regional Authorities (KS) (www.ks.no) has played a key role in the introduction of Lean in Norwegian municipalities. For example, KS was heavily involved in the pilot implementations in several Norwegian municipalities in 2010 (Ringen & Rolfsen, 2014). KS

has also functioned as a type of competence center, providing Norwegian municipalities with consulting and training services related to Lean. For example, KS's subsidiary KS-Konsulent (www.kskonsulent.no) provides consulting services related to Lean, and has developed a version of Lean ("K-Lean") which is tailored to municipalities.²

In addition, Norwegian municipalities have been inspired by and learned from Danish municipalities. To this point, it has been noted that Danish consultants were involved in some Norwegian Lean implementation projects in the early phase (Ringen & Rolfsen, 2014, p. 121).

It can be argued that Lean is a new phenomenon in the Norwegian municipality sector, since pilots and experimentation with Lean took place as late as in 2010 (Ringen & Rolfsen, 2014). In addition, the two aforementioned surveys of Norwegian municipalities conducted in the early 2010s, documented that Lean is a new phenomenon in Norwegian municipalities, but that these organizations relatively quickly starting learning about and experimenting with the Lean concept (Dolva, 2011; Schie, 2012).

Overall, the review of extant related research on the use and application of Lean in Scandinavian municipalities suggests that the Lean concept has had different trajectories in each of the three Scandinavian countries. Furthermore, relatively little is known about the diffusion and use of Lean in municipalities in Norway. Therefore, a survey of Norwegian municipalities could provide an updated picture of the concept's diffusion and application and could help identify and shed light on recent developments and trends.

3. Methods and data

This section outlines and describes the research approach followed in this study, focusing on issues related to overall research approach, questionnaire design, and response rate.

3.1. Electronic survey methodology

The study employs an electronic survey (e-survey) methodology. In the case of this study, the survey was administered using the survey software MI Pro (www.mipro.net), and the data were exported from MI Pro to the statistics software SPSS for further analysis.

All of the Norwegian municipalities' email addresses were gathered from a website (www.kommuneforlaget.no), and thereafter, an email invitation to take part in the survey was sent to each municipality's public email address. The email invitation contained a brief explanation of the purpose of the study, as well as a link to the web survey.

The invitation contained an explicit instruction for the recipient to forward the email to the person best suited to answer a survey about Lean. In this study, the aim was to target individuals with relevant managerial roles in the municipality, such as general managers, financial managers, or internal Lean consultants/advisors who would be able to answer questions about the use of Lean in their respective municipality in a meaningful way.

3.2. Questionnaire design

As discussed in the literature review, two previous surveys of Norwegian municipalities were conducted by Dolva (2011) and Schie (2012). In addition, another broad-based survey of Norwegian organizations was conducted in 2015 (Madsen et al., 2016). Many of the questions used in the current survey build on those used in these aforementioned surveys. In part, this makes it possible to make comparisons and to identify developments and trends in the adoption, diffusion and implementation of Lean in Norwegian municipalities.

At the same time, it should be noted that no particular test of the questions' reliability and validity has been carried out. However, the questions are not supposed to reflect complex, underlying concepts that need to be measured by multiple questions (for instance by employing a reflective

Table 1. Response rate			
	Frequency	Percentage	
Complete	109	25.55	
Answered via email (declined)	3	0.7	
Incomplete	16	3.75	
Non-response	298	70.0	
Sum	426	100.0	

measurement model). For most questions there is a simple one-to-one relationship between the question and the underlying concept. This makes lack of validity less of a concern. The questions have also been employed in the three aforementioned surveys (Dolva, 2011; Madsen et al., 2016; Schie, 2012); all of which have provided consistent and reliable results.

The questionnaire consisted of a mix of closed-ended and open-ended questions. However, most of the questions were closed-ended, and used five-point Likert scales. For many of the questions, it was possible for the respondents to select several answer alternatives. This means that in some of the tables, the sum of the percentages add up to more than 100 percent. Moreover, answer alternatives, such as "do not know" and "do not wish to answer" as well as open-ended answer alternatives to some of the questions were included. The purpose of this was to avoid forcing the respondents to select an answer alternative they did not feel appropriate. However, the analysis of the results in Section 4 excludes open-ended answers as few respondents provided such answers.

3.3. Response rate

The survey was sent to all 426 municipalities in Norway in February 2017. After two reminders that were sent out at two-week intervals, the final response rate ended at roughly 25 percent (Table 1). While this response rate may be considered relatively low judging by conventional standards in the research methods literature (see e.g. Ghauri & Grønhaug, 2002; Mitchell & Jolley, 2012), studies suggest that response rates in web and internet-based surveys tend to be lower (Cook, Heath, & Thompson, 2000; Shih & Xitao Fan, 2008). The fact that about 74 percent did not respond to or complete the survey, suggests that there could be potential issues related to non-response bias (Armstrong & Overton, 1977), which could represent a threat to the validity and generalizability of the findings.

4. Findings

This section consists of three main parts, each providing results about different aspects of the diffusion and use of Lean (adoption, implementation, and effects).

4.1. Adoption

This section reports on findings related to the adoption of the Lean concept in Norwegian municipalities. The section starts off with findings related to awareness and contact points, before examining adoption rate, time of adoption, adoption motives, reasons for non-adoption, and finally, planned adoption.

4.1.1. Awareness

The first issue concerns the awareness of Lean among Norwegian municipalities. It is important to examine the level of awareness, as municipalities are unlikely to adopt management concepts that they have not heard of. Table 2 shows that the awareness of Lean has increased sharply since the early 2010s. While the surveys by Dolva (2011) and Schie (2012) both showed that nearly 60 percent reported having heard of Lean, in 2017 nearly 92 percent report having heard of the concept.

This level of awareness is close to the level reported in a broad-based study of Lean adoption and diffusion in Norway (Madsen et al., 2016). The overall increase in the awareness of Lean over time is

Table	Table 2. Awareness of Lean in Norwegian municipalities 2011–2017						
	2011 (Dolva, <mark>2011</mark>)		2012 (Schie, 201	2012 (Schie, <mark>2012</mark>)		dy)	
	N	%	N	%	N	%	
Yes	108	58.7	110	58.5	100	91.7	
No	76	41.3	78	41.5	9	8.3	
Sum	184	100.0	188	100.0	109	100.0	

not surprising given the attention that has been devoted to Lean by Norwegian government agencies, politicians, and other actors (e.g. consulting firms) (Aspøy, 2014; Rolfsen, 2014). For example, Madsen et al. (2016) showed that there was a sharp increase in print-media discourse about Lean in Norway in the period from 2010–2016.

4.1.2. Contact points

The second issue is related to how municipalities have encountered the Lean concept, i.e. what the contact points are. Table 3 shows the municipalities' contact points with the Lean concept. The data indicate that the most important contact points were educational programs, other municipalities, previous employers, conferences, and employees.

It is a bit surprising that only three percent of the respondents point to consultants as a contact point since consultants play a leading role in the diffusion and popularization of management concepts and ideas (Heusinkveld, 2013; Jung & Kieser, 2012). The percentage mentioning consultants as a contact point is also much lower than what was reported in the study by Madsen et al. (2016). This rather surprising finding can be explained in different ways. One explanation for this difference in the role of consultants could perhaps be attributed to a lower degree of usage of consulting services among municipalities than among private sector firms. However, another explanation could be that the survey respondents who were forwarded the survey within the municipalities and may be responsible for Lean implementation, were not the ones who were in contact with consultants in the adoption phase (e.g. CEOs or³ mayors).

Table 4 shows that 24 percent of respondents report that they have not participated in any courses about Lean at all. More than half of the respondents report that their employees have participated, while roughly a third report that managers have participated. Twenty percent report that union representatives have participated in courses about Lean. The involvement of union

	Frequency	Percentage
Employees	11	11.0
Media	7	7.0
Other municipalities	17	17.0
Educational programs	18	18.0
Previous employer	16	16.0
Courses/conferences arranged by private sector actors	0	0
Courses/conferences arranged by public sector actors	14	14.0
Courses/conferences arranged by KS	7	7.0
Representatives from KS	3	3.0
Meetings with KS	1	1.0
Projects run by KS	0	0.0
Consulting firms	3	3.0
Other	3	3.0

able 3. Contact points with Lean (N = 100)

Table 4. Participation in courses about Lean (N = 100)			
Have representatives from the municipality participated in courses about Lean?	Frequency	Percentage	
Managers	35	35.0	
Employees	54	54.0	
Union representatives	20	20.0	
Do not know/unsure	16	16.0	
No	24	24.0	

representatives in seeking out information about Lean is not surprising since researchers have noted that unions have been actively involved in the introduction of Lean in Norway (Rolfsen & Ingvaldsen, 2012).

The respondents were also asked about the types of actors organizing and arranging these Leanrelated courses and seminars. Table 5 shows that many respondents reported that they did not know who organized and arranged the courses. Among those respondents who were able to recollect this, the organizations KS and Lean Forum Norge (Lean Forum Norway) were most commonly mentioned as having organized and arranged Lean courses. Moreover, some respondents identified other consulting, research, and analyst firms (e.g. Flowit AS, Ikaro Solutions AS, SINTEF, Lean Consulting AS, and Bedriftskompetanse AS). Most of these firms have been actively involved in the Norwegian market for Lean consulting services, e.g. by being listed in the "Lean consulting guide" on Lean Forum Norge's website.⁴

4.1.3. Contact with other municipalities

The next issue concerns contact with other municipalities, since management concepts and practices may spread via inter-organizational networks and ties (Abrahamson, 1991; DiMaggio & Powell, 1983). In a related question, the respondents were asked about whether the municipality had been in contact with other municipalities. Table 6 shows that about 60 percent have not been in contact with other municipalities, while about 40 percent agree that they have at least partly been influenced by, or themselves influenced, other municipalities. Furthermore, 33 percent report that they have visited other municipalities that are current Lean users, while 21 percent report that other municipalities have visited them.

Following up on this question, the respondents were asked about the extent to which other municipalities' adoption and experiences with Lean has influenced them. Table 7 shows that about 40 percent agree (at least partly) that other municipalities' adoption of Lean has been an influential factor in their own choice of whether or not to adopt Lean. Sixty percent agree that they have derived utility from other municipalities' knowledge of Lean, suggesting that a degree of inter-organizational learning has taken place.

Table 5. The actors arranging courses about Lean (N = 60)		
	Frequency	Percentage
KPMG	1	1.7
PwC	1	1.7
Lean Lab	1	1.7
Lean Forum Norge	6	10.0
KS⁰	17	28.3
Do not know	22	36.7
Other	12	20.0
	60	100.0

°KS was not an answer alternative in the original questionnaire, but added as a separate category in the table.

Table 6. Contact with other municipalities (N = 100)				
	Frequency	Percentage		
We have visited other municipalities using Lean	33	33.0		
Other Lean-using municipalities have visited us	21	21.0		
No	61	61.0		

Table 7. Influence by other municipalities (N = 27)				
	Other municipalities' adoption of Lean has been decisive in our choice	We have derived utility from other municipalities' knowledge of Lean		
Disagree completely (%)	29.6	25.9		
Disagree partly (%)	7.4	3.7		
Neither agree nor disagree (%)	22.2	7.4		
Agree partly (%)	25.9	22.2		
Agree completely (%)	14.8	40.7		
Do not know (%)	0.0	0.0		
Sum	100.0	100.0		

4.1.4. Adoption rate

In order to say something about the diffusion of Lean, it is important to examine the adoption rate, i.e. the percentage of municipalities that have adopted Lean. The surveys by Dolva (2011) and Schie (2012), which were carried out relatively soon after the emergence of Lean in the Norwegian municipality sector, documented adoption rates of 12 and about 18 percent, respectively. Table 8 shows that, roughly 5 years later, 33 percent claim to have adopted Lean. This means that the adoption rate has increased considerably over course of the last six years, and is now at a fairly high level. It is also worth noting from Table 8 that three percent of Norwegian municipalities have already abandoned Lean. The respondents were not asked a follow-up question about why they have abandoned Lean. However, as shown in Section 4.3.4, many respondents report having experienced implementation challenges. It is therefore possible that some may have dropped the Lean concept because of such implementation difficulties.

4.1.5. Time of adoption

Related to the previous question, the respondents were asked about when they had adopted Lean. Table 9 shows that almost half of the municipalities are very recent adopters of Lean, having adopted the concept within the last two years. Less than a quarter of the respondents answered that they adopted Lean more than five years ago (i.e. prior to 2012). Overall, these data support the view that Lean is still a relatively new phenomenon in the Norwegian municipality sector.

Table 8. Adoption rates in studies of Lean in Norwegian municipalities						
	2011 (Dolva, <mark>2011</mark>)		2012 (Schie, <mark>2012</mark>)		2017 (Current study)	
	N	%	N	%	N	%
Yes	13	12.0	20	18.2	33	33.0
No	95	88.0	90	81.8	64	64.0
Used Lean previously, but have abandoned it	N/A		N/A		3	3.0
Sum	108	100.0	110	100.0	100	100.0

Table 9. Time of adoption (N = 33)				
	Frequency	Percentage		
Less than 1 year ago	9	27.3		
1-2 years ago	7	21.2		
3-4 years ago	9	27.3		
5–10 years ago	8	24.2		
Do not know	0	0.0		
Sum	33	100.0		

4.1.6. Adoption motives

Related to adoption, another important issue pertains to the motives and rationales for adopting new management concepts (Abrahamson, 1991; Sturdy, 2004). Therefore, one question focused on the municipality's motives and rationales for adopting Lean. The answer categories that are shown in Table 10 are based on the influential typology of motives for adoption of management fads and fashions introduced by Abrahamson (1991).

Table 10 shows that the most commonly cited rationale for adopting Lean is that the concept is viewed as the most effective management tool, i.e. an efficiency-driven motive. About 11 percent answer pressure from politicians and the government, which is similar to what Abrahamson (1991) labels "forced selection." Nearly 19 percent report that others have recommended Lean ("fashion"), while about 7 percent report that other municipalities are using Lean ("fad"). Compared with the survey by Madsen et al. (2016), motives related to fashion, fad and forced-selection seem to play a bigger role in the municipality sector than in other sectors in Norway.

4.1.7. Reasons for non-adoption

In the previous questions, respondents were asked about the motives for adoption of Lean. Related to this, it is important to find out why non-adopters of Lean have chosen not to adopt the concept. Therefore, the respondents were asked about reasons for why they had not adopted Lean. As shown in Table 11, the respondents cited several different reasons. The most common claim was that they were using other management tools, that the effects of Lean were unclear, or that Lean was too resource-intensive.

These findings suggest that other management tools currently have a strong position in the Norwegian municipality sector. Recent research lend some support to this view, by showing that Norwegian municipalities are frequent users of other management concepts, such as the Balanced

	Abrahamson (1991) typology	Frequency	Percentage
Other municipalities are using Lean	Fad	2 (1)	7.4
Lean is the most effective management tool	Efficient choice	12 (4)	44.4
Others have recommended Lean	Fashion	5 (4)	18.5
Pressure from politicians and the government	Forced selection	3 (1)	11.1
Do not know		1	3.7
Other		4 (16)	14.8
		27	100.0

^oThe numbers in the parentheses are the respondents' original answers. As can be seen from the Table, 16 respondents choose to answer "other." However, after careful reading and analysis, we determined that twelve of these answers were closely related to the four categories in Abrahamson's typology. For example, eight respondents answered something which was closely related to an efficiency-related motive.

	Frequency	Percentage
Using other management tools	20	28.6
Change fatigue	0	0.0
Do not need it	2	2.9
Too resource intensive	8	11.4
Not feasible due to financial reasons	0	0.0
Effects are unclear	15	21.4
Do not know	14	20.0
Other	11	15.7
	70	100.0

Scorecard (Fallan, Olsen, Daleq, & Hobbel, 2015) and benchmarking (Askim, Johnsen, & Christophersen, 2008). It could be that municipalities may be less inclined to try Lean when the effects are perceived to be unclear. After all, managers are unlikely to adopt new management concepts unless they perceive potential benefits (Benders, 1999).

4.1.8. Non-adopters' use of other management concepts and tools

Those respondents who answered that they had either not heard of or adopted Lean, were asked about what types of management concepts and tools they use. Table 12 shows that nearly 90 percent used performance-oriented budgets, whereas balanced scorecards were used by roughly a third of the municipalities.

4.1.9. Respondents considering Lean adoption

Those municipalities that had not adopted Lean were asked about whether the municipality is considering future adoption of Lean. About 29 percent of the non-adopters were considering adoption (28.6%, N = 70), 23 percent reported that they did not know (22.9%, N = 70), whereas nearly 50 percent answered that they were not considering adopting Lean (48.6%, N = 70). Compared with the findings of Madsen et al. (2016), the percentage of municipalities considering adoption is relatively high. This could cautiously be interpreted as a sign that the adoption rate could climb further in the future.

4.2. Implementation

This section focuses on the part of the survey that focuses on how the Lean concept has been implemented in Norwegian municipalities. This entails a description of how the municipalities have contextualized the Lean concept, e.g. in terms of implementation in different functional areas. Furthermore, the section examines aspects of the implementation process, such as the degree to which employees and union representatives have participated, what types of Lean tools and techniques are used, and whether external consultants have been involved. Finally, the section contains related issues, such as perceptions of the degree to which Lean is implemented, and as well as whether or not Lean is used alongside other management concepts and tools.

Table 12. Non-adopters use of other management concepts and tools (N = 82)			
	Frequency	Percentage	
Performance-oriented budgets	73	89.0	
Balanced scorecard	27	32.9	
Activity based costing/management	0	0.0	
Benchmarking	15	18.3	
No management tool	0	0.0	
Do not know	5	6.1	
Other	6	7.3	

4.2.1. Contextualization of Lean

Contextualization refers to how a management concept, such as Lean is adapted, customized and translated to fit in a particular organization (see e.g., Røvik, 2007). As noted by several researchers (Benders & van Bijsterveld, 2000; Brännmark et al., 2012; Morris & Lancaster, 2006; Wittrock, 2015, 2017), the Lean concept is vague and open to interpretation, which means that the Lean concept is likely to be contextualized and translated in different ways as it is implemented in organizational practice.

4.2.1.1. Lean project manager: In Lean implementation projects it is typically important to have an internal Lean "navigator" or change agent (Breit & Rolfsen, 2014b; Rolfsen, 2014). This person may be an individual with knowledge and experience with Lean, and is given the task of driving and developing the Lean concept within the organization. Therefore, the respondents were asked about whether the municipality has a Lean project manager. Nearly half of the respondents (44.4%, N = 27) answer that they have a designated Lean project manager.

4.2.1.2. Choice and adaptation of the Lean concept: As mentioned earlier, the Lean concept has considerable room for interpretation and is implemented and translated in different ways (Benders & van Bijsterveld, 2000; Brännmark et al., 2012; Morris & Lancaster, 2006; Wittrock, 2015). The survey data also suggest that the municipalities vary in terms of how they interpret and apply the Lean concept. Table 13 shows that more than half of the municipalities claim to have developed their own "home-made" Lean model. About 15 percent have adopted a Lean model from another municipality, whereas about 25 percent have adopted a Lean model from external consultants.

4.2.1.3. Evolution of the Lean concept after adoption and implementation: In the literature on management concept and ideas, it is suggested that concepts may take different trajectories in organizations (Røvik, 2007, 2011). Therefore, the respondents were asked about whether the municipality's Lean model has evolved since the Lean concept was initially introduced in the municipality. The data show that about three out of four (74.1%, N = 27) claim that their Lean model has evolved over time.

4.2.2. Implementation in different functional areas

Next, the respondents were asked about the implementation of Lean within different functional areas of the municipality. The respondents were asked to identify areas where Lean was used at the time of initial adoption/implementation, as well as where it is currently used. Table 14 shows that, initially, Lean was most commonly used within areas, such as health care, administration, and real estate and buildings. The table also reveals some changes between use at the time of adoption and current use. In particular, the use of Lean in areas, such as child care services, schools and education, real estate and buildings, and taxation and commerce has increased sharply. One possible explanation for the heavy use of Lean in these areas could be that these are big budget items for Norwegian municipalities.⁵ Therefore, municipalities may be particularly motivated to adopt and implement Lean in these areas.

4.2.3. Employee participation in the implementation process

Next, the focus turns to an examination of the perceived level of employee participation in the implementation process. Table 15 shows that there generally is a very high degree of employee participation in the Lean implementation process. This suggests that the implementation of Lean at least

Table 13. Municipalities' choice of what Lean model to adopt and implement (N = 27)					
	Frequency	Percentage			
Developed own Lean model	15	55.6			
Adopted a Lean model from another municipality	4	14.8			
Adopted a Lean model from external consultants	8	25.9			
Do not know	1	3.7			
Sum	27	100.0			

	At the time of initial adoption/implementation (%)	Current (%)
Health care	81.5	88.9
Administration	70.4	77.8
Real estate and buildings	55.6	74.1
Nature, culture, and leisure	14.8	25.9
Water and sanitation	14.8	25.9
Child care services	25.9	51.9
Schools and education	22.2	37.0
Taxation and commerce	3.7	18.5
Waste and recycling	7.4	7.4
Housing and social services	14.8	33.3
Fire department	3.7	11.1
Transportation, parking, and roads	7.4	14.8
Agriculture	3.7	N/Aª
Other	18.5	14.8

°Agriculture was mistakenly omitted as an answer alternative.

Table 15. Perceived level of employee participation in the implementation process (N = 27)						
	Disagree completely (%)	Disagree partly (%)	Neither agree nor disagree (%)	Agree partly (%)	Agree completely (%)	Do not know (%)
Employees had a high level of influence on the implementation of Lean	0.0	11.1	11.1	18.5	55.6	3.7
Employees participated in the implementation of Lean	7.4	7.4	18.5	25.9	40.7	0.0
Management are visible and active in the implementation of Lean	0.0	0.0	14.8	37.0	48.1	0.0
There are regular meetings where employees can suggest new solutions	0.0	18.5	14.8	18.5	40.7	7.4
Suggestions from employees are often taken into account	0.0	0.0	3.7	37.0	55.6	3.7

partly is a bottom-up process. These findings are not very surprising given that Lean in Norway typically has a more "democratic" flavor than in other countries (Ingvaldsen, 2013; Ingvaldsen et al., 2012; Rolfsen, 2014).

4.2.4. Union participation in the implementation process

A distinguishing feature of the implementation of Lean in Norway is that unions have been actively involved in the process (Ingvaldsen et al., 2012; Rolfsen, 2014; Rolfsen & Ingvaldsen, 2012). Therefore, the respondents were asked about how they perceived the level of union participation in the implementation of Lean. Table 16 shows that there is generally moderately high level of participation by union representatives in the Lean implementation process.

Table 16. Perceived union participation in the implementation of Lean (N = 27)						
	Disagree completely (%)	Disagree partly (%)	Neither agree nor disagree (%)	Agree partly (%)	Agree completely (%)	Do not know (%)
Union representatives had a high level of influence on the implementation of Lean	11.1	3.7	37.0	29.6	18.5	0.0
Union representatives participated in the implementation of Lean	18.5	18.5	11.1	33.3	14.8	3.7
Union representatives are visible and active in the implementation of Lean	11.1	18.5	25.9	29.6	14.8	0.0
There are regular meetings where union representa- tives can suggest new solutions	3.7	22.2	18.5	33.3	22.2	0.0
Suggestions from union representatives are often taken into account	3.7	7.4	29.6	37.0	14.8	7.2

4.2.5. Use of Lean-related tools and techniques

The "Lean toolbox" consists of a large number of tools and techniques (Bicheno, 2008; Bicheno & Holweg, 2008; Modig & Åhlström, 2012; Rolfsen, 2014). Organizations draw on and utilize these tools and techniques to varying degrees in actual implementations of Lean. Table 17 provides an overview of the most commonly used Lean-related tools and techniques among Norwegian municipalities. The six most widely used tools are (1) continuous improvement, (2) visual cards, (3) identification and removal of constraints, (4) 5S, (5) value stream mapping, and (6) five whys. The data show a pattern that to some extent resembles what has been seen in studies of Danish and Swedish municipalities (Arlbjørn & Freytag, 2009; Brännmark, 2012; Stentoft Arlbjørn et al., 2011).

	Frequency	Percentage
/alue stream mapping	20	74.1
Five whys	19	70.4
Continuous improvement	25	92.6
Identification of waste	19	70.4
Kanban	2	7.4
SMED	0	0.0
Visual cards	25	92.6
55	20	74.1
Six Sigma	0	0.0
Identification and removal of constraints	21	77.8
A3	11	40.7
Systematic observation of work processes	9	33.3
PDCA	14	51.9
Standardized work processes	14	51.9
Poka-Yoke	1	3.7
Quality circles	7	25.9
Other	4	14.8

For example, in Denmark, Arlbjørn and Freytag (2009) found that tools, such as continuous improvement, value stream mapping, and identification of waste were widely used. Similarly, in Sweden, value stream mapping, 5S, identification of waste, as well as identification and removal constraints were much used (Brännmark, 2012; cited in Ringen & Rolfsen, 2014).

4.2.6. Use of external consultants

The respondents were asked about the extent to which external consultants have been involved in the implementation of Lean in their municipality. This question stems from previous research on Lean in Norway which has shown that external consultants have been involved in the implementation of Lean in many organizations (Breit & Rolfsen, 2014a). For example, it has been reported that consultants were actively involved in some of the early initiatives in the Norwegian municipality sector (Ringen & Rolfsen, 2014).

Table 18 shows that consultants generally have played a very active role in the adoption and implementation of Lean in Norwegian municipalities. Only about 15 percent answer that consultants have not been involved. Nearly half of the respondents answer that external consultants have been involved in relation to top or middle managers. Earlier in this paper, it was shown that only three percent stated that consultants were the point of contact with Lean. An explanation for this difference is that some municipalities may have hired external consultants later in the implementation process.

Following up on the previous question, the respondents were asked if the municipality kept using consultants in the post-adoption phase. Table 19 shows that most of the municipalities have not used consultants to a great extent in the post-adoption phase. While Table 18 above indicates that consultants have played an influential role in the early adoption phase, the data in Table 19 suggest that these consultants have mostly left the municipalities after the initial development and roll-out of Lean.

4.2.7. Degree of Lean implementation

Related to contextualization, it is interesting to find out how the respondents perceive the degree to which Lean is implemented in the municipality. Therefore, the respondents were asked to indicate their perception of the degree to which Lean is implemented in the municipality. Table 20 shows that

Table 18. Use of external consultants (N = 27)		
	Frequency	Percentage
External consultants have developed a ready-to-use solution	3	11.1
External consultants have been involved in relation to top managers	13	48.1
External consultants have been involved in relation to middle managers	12	44.4
External consultants have been involved in relation to union representatives	6	22.2
External consultants have been involved in relation to other employees	8	29.6
No	4	14.8
Do not know	1	3.7
Other	5	18.5

Table 19. Use of external consultants post-adoption (N = 27)					
	Frequency	Percentage			
Yes	2	7.4			
Partly	7	25.9			
No	18	66.7			
	27	100.0			

Table 20. Degree of implementation (N = 27)					
	Frequency	Percentage			
0–20%	10	37.0			
20-40%	7	25.9			
40-60%	6	22.2			
60-80%	3	11.1			
80-100%	0	0.0			
All units	1	3.7			
Do not know	0	0.0			
	27	100.0			

most respondents perceive a relatively low level of Lean implementation. More than a third perceive that they currently are less than 20 percent Lean. Only a single respondent claims to be "completely" Lean. These findings can be seen in relation to the findings in Section 4.1.5 showing that most municipalities have recently adopted Lean. Hence, they may not have been able to venture far on their "Lean journeys." After all, in the Lean literature it is highlighted that the implementation of Lean could take several years (e.g. Womack & Jones, 2003).

4.2.8. Use of other management concepts in addition to Lean

It is also interesting to find out what other types of management concepts municipalities used in addition to Lean. Researchers have shown that new management concepts are often combined with other existing concepts and practices (Heusinkveld & Benders, 2012; Røvik, 2011). Table 21 shows that a majority of the respondents reported that they use budgets and balanced scorecards, while benchmarking is used to a lesser extent. The respondents who answered "other" claimed that they used different types of causal analyses, priority matrices or "home-made" variants of Lean-related tools and techniques.

4.3. Effects

This section turns the focus to what Norwegian municipalities perceive as the effects of Lean implementation. This involves a description of what the aim of implementing Lean was, whether or not the municipality has evaluated effects, the Lean concept's fit with organizational culture, as well as the challenges associated with the implementation of Lean.

4.3.1. Aim of implementation

The respondents were asked about the aim of implementing Lean (Table 22). The results show that, not surprisingly, most respondents claimed that key aims of implementing Lean were to improve resource utilization and productivity, organizational development, or key user or employee satisfaction measures. It is noteworthy that none of the respondents referred to "downsizing" as being the aim of Lean implementation, since Lean is often strongly linked to rationalization efforts, such as downsizing (Benders & van Bijsterveld, 2000). However, the finding that none are using it for rationalization could be explained by the fact that Lean is often implemented in a more democratic and

Table 21. Use of other management tools and techniques in conjunction with Lean (N = 27)					
	Frequency	Percentage			
Budgets	24	88.9			
Balanced scorecard	17	63.0			
Activity based costing/management	0	0.0			
Benchmarking	5	18.5			
Do not know	1	3.7			
Other	3	11.1			

	Frequency	Percentage
Improve user/customer satisfaction	23	85.2
Improve resource utilization/productivity	26	96.3
Improve employee satisfaction	23	85.2
Organizational development	23	85.2
Downsizing	0	0.0
Do not know/do not wish to answer	0	0.0
Other	2	7.4

cooperative way in Norwegian work life, which has a strong focus on employee rights and powerful unions (Rolfsen, 2014; Rolfsen & Ingvaldsen, 2012).

4.3.2. Evaluation of effects

Related to the question of what was the aim of implementing Lean, is finding out whether the municipalities have evaluated the effects of Lean implementation. More than half of the respondents (51.9%, N = 27) indicate that they have evaluated the effects of implementation. The 14 respondents who answered that they have evaluated the effects of Lean implementation, were asked another question about the perceived effects of Lean implementation on different performance areas. Table 23 shows that the respondents predominantly perceive positive effects. This is similar to the pattern documented by Madsen et al. (2016). However, a large portion of the municipalities report that Lean is too recently implemented to be able to evaluate effects.

4.3.3. Perceptions of fit with organizational culture

The respondents who had evaluated effects were also asked about how Lean is perceived to fit with the organization's culture, and how the concept has been received by employees. Table 24 shows that Lean is generally perceived as compatible with the organizational culture. Furthermore, the concept is perceived to be well-received by employees. Generally, these findings suggest that respondents have positive evaluations of the effects of the concept.

4.3.4. Implementation challenges

Research on Lean suggest that there are many issues and challenges associated with the implementation of the concept (Bhasin & Burcher, 2006; Krause-Jensen, 2017; Loenen & Schouteten, 2016; Stentoft Arlbjørn & Vagn Freytag, 2013; Yadav, Nepal, Rahaman, & Lal, 2017). Therefore, one survey question asked the respondents about whether the municipality has encountered challenges in the implementation process. Table 25 shows that the most commonly cited problems are cultural challenges, lack of top management engagement, resistance from employees, communication problems. However, 37 percent report no particular challenges.

Table 23. Perceiv	Table 23. Perceived effects of Lean implementation on different performance areas (N = 14)							
	Resource utilization	Productivity	Quality	User/customer satisfaction	Flexibility and response time	Employee satisfaction	Sickness leave	Finances
Very negative (%)	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Negative (%)	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
None (%)	0.0	7.1	0.0	7.1	0.0	14.3	21.4	7.1
Positive (%)	71.4	50.0	71.4	35.7	50.0	42.9	35.7	42.9
Very positive (%)	21.4	21.4	14.3	21.4	21.4	7.1	0.0	14.3
Too recently implemented (%)	7.1	21.4	14.3	14.3	14.3	21.4	14.3	14.3
Do not know (%)	0.0		0.0	21.4	14.3	14.3	28.6	21.4

Table 24. Perceptions of the Lean concept's fit with the municipality's organizational culture (N = 14)

	Lean is compatible with the organizational culture	Lean has been well-received by employees
Disagree completely (%)	0.0	0.0
Disagree partly (%)	0.0	0.0
Neither agree nor disagree (%)	11.1	11.1
Agree partly (%)	37.0	70.4
Agree completely (%)	51.9	18.5
Do not know (%)	0.0	0.0
	100.0	100.0

Table 25. Implementation challenges (N = 27)		
	Frequency	Percentage
Resistance from employees	10	37.0
Resistance from union representatives	3	11.1
Cultural challenges	15	55.6
Lack of top management engagement	14	51.9
Communication problems	7	25.9
No particular challenges	10	37.0
Do not know	0	0.0
Other	1	3.7

4.3.5. Evaluation of the future of the Lean concept

Finally, the respondents were asked a question concerning their evaluation of the future of the Lean concept. Table 26 shows that most respondents tend to disagree that Lean is a passing trend. This is can be viewed an indication of a relatively strong faith in the future of Lean as a management concept, suggesting that the concept is currently highly institutionalized in the Norwegian municipality sector. It seems that few respondents are showing signs of being disillusioned with the concept or that Lean is becoming "worn out" (Benders & Van Veen, 2001). However, in other contexts, Lean has experienced downturn periods (Benders & van Bijsterveld, 2000; Larsson, 2012; Wittrock, 2015). Therefore, it is possible that this could happen in the Norwegian municipality sector as well at some point in the future.

4.4. Summary of results

The data about the adoption of Lean have shown that the awareness of the concept has increased quite sharply since the early 2010s. The survey responses indicate that municipalities have come in contact with Lean in different ways, such as through participation in educational programs, conferences and contact with other municipalities. As a result, the adoption rate has increased to about

Table 26. Evaluation of the future of the Lean concept ($N = 27$)		
	Lean is a passing trend	Lean is an effective tool for reducing waste
Disagree completely (%)	66.7	11.1
Disagree partly (%)	18.5	0.0
Neither disagree nor agree (%)	11.1	0.0
Agree partly (%)	3.7	29.6
Agree completely (%)	0.0	59.3

one third. However, most municipalities are relatively recent adopters. While most claim that they made the decision to adopt Lean because it is an efficient management tool, others report that they have been influenced by other municipalities, fashion-setters, or have been pressured by politicians and the government. Moreover, there are still relatively many municipalities considering adopting Lean, which can cautiously be interpreted as a sign that the adoption rate in the municipality sector could continue to climb.

The main issue in the part about the implementation of Lean has been how Lean is contextualized within the municipalities, i.e. how the concept is adapted and customized. In the survey, the respondents reported that Lean is most commonly used within areas, such as health care, administration, real estate and buildings, and child care services. The survey also reveals interesting findings about the implementation process. For example, employees and union representatives have been actively involved. The municipalities also draw on Lean-related tools and techniques to varying extent. Overall, the respondents perceive a relatively low degree of Lean implementation, which could be attributed to the relatively recent adoption of Lean. Finally, the results show that the municipalities use other management tools and techniques (e.g. budgets, balanced scorecards) in conjunction with Lean.

Finally, the part about the effects of Lean shows that the main aims of Lean adoption and implementation are related to increasing productivity and improving customer/employee satisfaction. While the respondents generally have a very favorable view of the effects of Lean, many also report implementation challenges related to culture, managerial engagement, and organizational resistance. Most respondents tend to have a positive evaluation of the future of the Lean concept, generally disagreeing with the view that Lean is a passing trend, and agreeing that is as an effective tool for reducing waste.

5. Discussion

In this section the survey results are discussed in light of extant research on Lean in the municipality sector, and, more generally, the literature on the diffusion of popular management concepts and ideas. The discussion is divided into three parts: (1) adoption, (2) implementation, and (3) effects.

5.1. Adoption

The survey data have cast light on several aspects of the adoption process in Norwegian municipalities. With regards to the pre-adoption phase, i.e. where organizations learn about and become aware of a management concept (Madsen, 2014; Rossem & Veen, 2011), the survey data show that the awareness of Lean has increased sharply since 2011/2012, to the point where more than 90 percent are aware of the concept.

One explanation for this increase in awareness is that various types of fashion-setting actors (e.g. Lean consultants, gurus, and academics) have created a "wave of interest" (Kieser, 1997) around Lean. This view is partly supported by research which has shown that there has been a great increase in the media discourse about Lean in Norway in recent years, as well as a wide range of other Lean-related activities, such as conferences, seminars, training programs, and networks (Madsen et al., 2016).

Furthermore, the data suggest that municipalities, at least to a certain extent, appear to have been influenced by social and institutional factors. For example, municipalities report that they came in contact with the concept via interaction with different types of fashion-setting actors (e.g. conference/seminar organizers), or that they learned about the concept as a result of contact and interaction with other municipalities.

With regards to the adoption decision-making phase, the survey also sheds some light on motives and rationales driving the decision of whether or not to adopt Lean. The survey suggests that efficiency, fad, fashion, and forced-selection motives (Abrahamson, 1991) have been influential in the adoption decision-making process. While the survey data are not able to shed much light on this, it is possible that a combination of different motives have worked in interplay in the municipality sector (cf. Daniel, Myers, & Dixon, 2012; Ilmari Rautiainen, 2009; Sturdy, 2004). Overall, the data provide support for the role of institutional isomorphism in the adoption of new management concepts (DiMaggio & Powell, 1983), i.e. mimetic pressure ("other municipalities are using Lean"), coercive pressure ("pressure from politicians and the government"), and normative pressure ("others have recommended Lean").

Lastly, the data suggest that the adoption rate has increased considerably, and many of the municipalities are recent adopters, which suggests that Lean is still in the early stages of its lifecycle in the municipality sector in Norway. The lifecycle pattern in the municipality sector differs from the pattern in Norway as a whole where Lean appears to be in a relatively later stage of the lifecycle (Madsen et al., 2016). This indicates that the adoption and diffusion pattern of the Lean concept differs quite a bit between the private and public sectors in Norway.

5.2. Implementation

The findings also cast light on several aspects of the implementation of Lean in the Norwegian municipality sector. These findings have implications for discussions about the nature of the Lean concept, and how it is interpreted and implemented in organizational practice (Benders & van Bijsterveld, 2000; Brännmark et al., 2012; Wittrock, 2015). With the regards to implementation, the data show that Norwegian municipalities use of a wide range of Lean-related tools and techniques. Similar to the observations made in Denmark by Stentoft Arlbjørn et al. (2011), it appears that Norwegian municipalities draw on and utilize a wide variety of tools and techniques from the "Lean Toolbox," but it is not clear from this study whether municipalities use the different Lean tools and principles together in a "holistic" and integrated way.

The data also suggest that the Lean concept is often used in combination with other management concepts and tools (e.g. Balanced Scorecards), suggesting that "translation" (Røvik, 2007) and adaptation of the Lean concept takes place. The perceptions of a relatively low degree of implementation also suggest that interpreting and applying the Lean concept is time-consuming and challenging in many ways. Moreover, the high level of participation by employees and union representatives in Norwegian municipalities also highlight the fluid and negotiated nature of the Lean concept in the public sector (Rahbek Gjerdrum Pedersen & Huniche, 2011b). It is possible that the involvement of such organizational actors could influence the degree of implementation. For example, some employee and union representatives may be skeptical of Lean, and may argue for a more gradual and incremental increase in the use of Lean in order to be able to assess the impact on the work environment.

5.3. Effects

Finally, the survey data shed light on effects of adoption and implementation of the Lean concept. Overall, the respondents tend to be quite positive in their perceptions of the effects of Lean in their organizations. At the same time, they also point to different types of implementation challenges. Interestingly, this is much of the same pattern that was revealed in the survey by Madsen et al. (2016). As discussed previously, the Lean concept appears to be in the early stages of its life cycle in the Norwegian municipality sector, as most municipalities have adopted Lean relatively recently. Therefore, it is possible that Norwegian municipalities are still in a "honeymoon period," in which they are less likely to focus on negative aspects of implementing and using the Lean concept.

Another reason for the largely positive evaluations could be that few negative experiences or failure cases from the Norwegian municipality sector have surfaced in the media. This means that managers of municipalities, at least for the time being, view Lean in a primarily positive light and perhaps through rosy-colored glasses. However, this could change relatively quickly if Lean fails to live up to the high expectations that probably exist in the municipality sector. If this were to happen,

the Lean concept could get "worn out" (cf. Benders & Van Veen, 2001). In the context of Lean, Wittrock (2015) has shown that the Lean concept has experienced a downturn in some parts of the world. Only time will show whether Lean will sustain its popularity in the Norwegian municipality sector.

6. Conclusions

6.1. Contributions

This paper has examined the diffusion of Lean in the Norwegian municipality sector. In doing so, the aim of the paper was to make a contribution to the literature on the use of Lean in the municipality sector, and provide a new and updated picture of the diffusion of the Lean concept in Norwegian municipalities.

Moreover, the study's findings could also be of practical use to managers of Norwegian municipalities that are currently using, or considering adopting and implementing Lean. There are several lessons to be learned for managers of municipalities, in particular for those considering adopting or those in the early phases of implementing Lean.

The findings of the study suggest that Norwegian municipalities use Lean in different ways, and utilize different types of Lean tools and techniques. It is important to be aware that Lean is not easy to implement; after all, many respondents report challenges related to organizational culture, managerial engagement, or organizational resistance. Furthermore, most report that they have implemented Lean to a low or moderate extent, which suggests that the "Lean-ification" process is perceived to be a long journey.

Therefore, it is important that managers of municipalities devote sufficient time and resources to the Lean implementation project. For example, Lean should not be viewed as a panacea which will immediately solve all problems related to productivity and efficiency. Rather, managers need to prepare the municipality for the fact that the "Lean journey" might not be smooth and linear as portrayed in textbooks or presentations by Lean consultants and gurus.

6.2. Limitations

Like all studies, the current study has several limitations that should be considered carefully. First, the response rate is relatively low, and as a result, there is a risk that non-response bias could threaten the validity of the results. Moreover, there is a low number of responses per question, which means that it has not been possible to conduct any robust correlation or regression analyses. Instead, the analysis carried out in this paper has had to remain descriptive.

Second, the fact that the recipients of the email invitation were asked to forward the email to the person best suited to answer the survey, may have introduced a source of bias. For example, there could be differences in terms of the ways the survey was circulated ("passed around") within the municipality, or differences related to whether the initial recipient of the email invitation had sufficient knowledge about who would be best suited to answer the survey.

Third, the cross-sectional nature of the current study means that it only provides a snapshot of the Lean concept's diffusion in the municipality sector. Finally, while the survey method is useful for obtaining a picture of the diffusion of Lean in the municipality sector as a whole, it does not offer much "rich" in-depth insight into how the Lean concept is interpreted, implemented and "translated" in a particular municipality.

6.3. Future work

Keeping in mind the limitations, there are several ways in which the current study could be extended in future research efforts. First, this study has shown that municipalities tend to draw on and utilize a broad spectrum of Lean-related tools and techniques in practice. At the same time, it is unclear whether these tools and techniques are used in a "holistic" and integrated way (cf. Stentoft Arlbjørn et al., 2011). Therefore, future studies could focus more closely on obtaining a better understanding of how different elements of the "Lean Toolbox" are chosen, applied and combined by municipalities in practice. An interesting question pertains to whether there are certain combinations of tools and techniques that are more effective than others.

Another avenue for future research pertains to the effects of using Lean in the municipality sector. The study shows that there is much variation when it comes to perceptions of the effectiveness of Lean. An important question is therefore why some municipalities succeed and appear to have seemingly smooth "Lean journeys," while others encounter considerable implementation difficulties and, in some cases, ultimately fail and end up discarding the concept. Such in-depth investigations could generate insights that could be of interest and use to practitioners struggling to apply Lean in their own organizations. Moreover, insight into the main pitfalls associated with Lean implementation in the municipality sector could potentially help Lean users avoid similar problems in the future.

Finally, researchers should utilize a broader range of methodological approaches when studying Lean in the municipality sector. For example, researchers could carry out qualitative research interviews (Qu & Dumay, 2011) in one or more municipalities. Such studies could cast light on how Lean is interpreted, applied and "translated" in practice (Andersen & Røvik, 2015). Related to this, is a need for longitudinal studies where researchers follows the implementation of Lean over time. Such longitudinal studies could uncover various organizational processes that happen as the Lean concept is implemented, and could help shed light on how organizational actors shape the organization's "Lean thinking" over time (cf. Røvik, 2011). Longitudinal fieldwork could also provide more insight into the cultural and social implications of Lean implementation in municipalities (Krause-Jensen, 2017).

Acknowledgments

The authors are indebted to two anonymous reviewers for their many helpful comments and suggestions on the first version of the manuscript. Their suggestions have helped shape the paper into its final form.

Funding

The authors received no direct funding for this research.

Author details

Dag Øivind Madsen¹ E-mail: dagm@usn.no ORCID ID: http://orcid.org/0000-0001-8735-3332 Stig Risvik¹ E-mail: srisvi@online.no

Tonny Stenheim²

E-mail: tonny.stenheim@bi.no

- ¹ School of Business, University College of Southeast Norway, Bredalsveien 14, 3511 Hønefoss, Norway.
- ² Department of Accounting, Auditing and Business Analytics, BI Norwegian Business School, 0442 Oslo, Norway.

Citation information

Cite this article as: The diffusion of Lean in the Norwegian municipality sector: An exploratory survey, Dag Øivind Madsen, Stig Risvik & Tonny Stenheim, *Cogent Business & Management* (2017), 4: 1411067.

Notes

- 1. https://www.regjeringen.no/en/find-document/norwegian-official-reports/id1767/.
- http://www.kskonsulent.no/ledelse-og-organisasjon1/klean/.
- Thanks to one of the reviewers for suggesting this alternative explanation.

- 4. www.leanforumnorge.no/lean-torget/konsulenttorget.
- See, for example, data from Statistics Norway https://www.ssb.no/kommregnko or the following website with contains data about the budget profiles of Norwegian municipalities: http://www.kommuneprofilen.no/ Profil/Okonomi/BenchBy/okonomi_komm_utgiftstype_ by.aspx, web links accessed 15 November 2015.

References

- Abrahamson, E. (1991). Managerial fads and fashions: The diffusion and rejection of innovations. Academy of Management Review, 16(3), 586–612.
- Alarcón, L. (1997). Lean construction. Boca Raton: CRC Press. https://doi.org/10.4324/9780203345825
- Andersen, H., & Røvik, K. A. (2015). Lost in translation: A casestudy of the travel of lean thinking in a hospital. BMC Health Services Research, 15(1), 376. https://doi.org/10.1186/s12913-015-1081-z
- Arlbjørn, J. S., & Freytag, P. V. (2009). Positionering eller effektivisering via lean i den kommunale sektor? Ledelse & Erhvervsøkonomi, 74(3), 55–71.
- Armstrong, J. S., & Overton, T. S. (1977). Estimating nonresponse bias in mail surveys. *Journal of Marketing Research*, 14, 396–402. https://doi.org/10.2307/3150783
- Askim, J., Johnsen, Å., & Christophersen, K.-A. (2008). Factors behind organizational learning from benchmarking: Experiences from Norwegian municipal benchmarking networks. Journal of Public Administration Research and Theory, 18(2), 297–320.
- Aspøy, A. (2014). Lean tar over styringen. Stat & Styring, 4, 1. Retrieved from https://www.idunn.no/stat/2014/04/ lean_tar_over_styringen
- Bateman, N., Radnor, Z., & Glennon, R. (2018). Editorial: The landscape of Lean across public services. *Public Money & Management*, 38(1), 1–4. https://doi.org/10.1080/09540962.2018.1389482

- Benders, J. (1996). Leaving Lean? Recent changes in the production organization of some Japanese car plants. *Economic and Industrial Democracy*, 17(1), 9–38. https://doi.org/10.1177/0143831X96171002
- Benders, J. (1999). Tricks and trucks: A case study of organization concepts at work. The International Journal of Human Resource Management, 10(4), 624–637. https://doi.org/10.1080/095851999340305
- Benders, J., & van Bijsterveld, M. (2000). Leaning on lean: The reception of a management fashion in Germany. New Technology, Work and Employment, 15(1), 50–64. https://doi.org/10.1111/ntwe.2000.15.issue-1
- Benders, J., & Van Veen, K. (2001). What's in a fashion? Interpretive viability and management fashions *Organization*, 8(1), 33–53.
- Bergström, T., & Lapsley, I. (2017). On the elusive nature of modernisation in government. Statsvetenskaplig tidskrift, 119(1), 7–22.
- Bhasin, S., & Burcher, P. (2006). Lean viewed as a philosophy. Journal of Manufacturing Technology Management, 17(1), 56–72. https://doi.org/10.1108/17410380610639506
- Bhatia, N., & Drew, J. (2006). Applying lean production to the public sector. *The McKinsey Quarterly*, 3(1), 97–98.
- Bicheno, J. (2008). The Lean toolbox for service systems. Johannesburg: PICSIE books.
- Bicheno, J., & Holweg, M. (2008). The Lean toolbox: The essential guide to Lean transformation. Buckingham: PICSIE Books.
- Brännmark, M. (2011). Lean administration: En litteraturgenomgång av lean när konceptet implementeras i kommuner. Paper presented at the Forum för arbetslivsforsknings årliga konferens, Luleå. Retrieved from http://www.diva-portal.org/smash/get/ diva2:460018/FULLTEXT01.pdf
- Brännmark, M. (2012). Lean i kommuner och myndigheter: En översikt över existerande empirisk forskningslitteratur. Stockholm: Innovationsrådet.
- Brännmark, M., Langstrand, J., Johansson, S., Halvarsson, A., Abrahamsson, L., & Winkel, J. (2012). Researching Lean: Methodological implications of loose definitions. *Quality Innovation Prosperity*, 16(2), 35–48.
- Breit, E., & Rolfsen, M. (2014a). Bruk av konsulenter i utviklingen av lean. In M. Rolfsen (Ed.), *Lean blir norsk -Lean i den norske samarbeidsmodellen* (pp. 211–221). Bergen: Fagbokforlaget.
- Breit, E., & Rolfsen, M. (2014b). Implementering av lean. In M. Rolfsen (Ed.), Lean blir norsk - Lean i den norske samarbeidsmodellen (pp. 198–210). Bergen: Fagbokforlaget.
- Cook, C., Heath, F., & Thompson, R. L. (2000). A meta-analysis of response rates in web- or internet-based surveys. *Educational and Psychological Measurement*, 60(6), 821– 836. https://doi.org/10.1177/00131640021970934
- Daniel, E., Myers, A., & Dixon, K. (2012). Adoption rationales of new management practices. Journal of Business Research, 65(3), 371–380. https://doi.org/10.1016/j. jbusres.2011.06.033
- DiMaggio, P. J., & Powell, W. W. (1983). The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields. *American Sociological Review*, 48(2), 147–160. https://doi.org/10.2307/2095101
- Dolva, M. M. (2011). Implementering av lean i norske kommuner: Har norske kommuner implementert lean, eller vurderer de å implementere det? Kristiansand: University of Agder.
- Drotz, E. (2014). Lean in the public sector: Possibilities and limitations. Linköping: Linköping University Electronic Press.
- Esping-Andersen, G. (1990). The three worlds of welfare capitalism. Cambridge: Polity press Cambridge.

- Fallan, E., Olsen, T.-E., Daleq, B., & Hobbel, M. A. (2015). Bruk av balansert målstyring i norske kommuner. *Magma*, *18*(8), 62–75.
- Ghauri, P., & Grønhaug, K. (2002). *Research methods in business studies* (2nd ed.). Harlow: Prentice Hall Europe, Pearson Education Limited.
- Green-pedersen, C. (2002). New public management reforms of the Danish and Swedish Welfare States: The role of different social democratic responses. *Governance*, 15(2), 271–294. https://doi.org/10.1111/gove.2002.15.issue-2
- Hallström, K. T., & Thedvall, R. (2015). Managing administrative reform through language work. Implementing Lean in Swedish public sector organisations. Scandinavian Journal of Public Administration, 19(2), 89–108.
- Heusinkveld, S. (2013). The management idea Factory: Innovation and commodification in management consulting. New York, NY: Routledge.
- Heusinkveld, S., & Benders, J. (2012). On sedimentation in management fashion: An institutional perspective. Journal of Organizational Change Management, 25(1), 121–142. https://doi.org/10.1108/09534811211199628
- Holmemo, M. D. Q., & Ingvaldsen, J. A. (2018). Local adaption and central confusion: Decentralized strategies for public service Lean implementation. *Public Money & Management*, 38(1), 13–20. https://doi.org/10.1080/09540 962.2018.1389493
- Holmemo, M. D. Q., Ingvaldsen, J. A., & Benders, J. (2017). Debate: Changing to Lean public services. *Public Money & Management*, 37(1), 5–6. https://doi.org/10.1080/0954096 2.2016.1249221
- Hood, C. (1995). The "new public management" in the 1980s: Variations on a theme. Accounting, Organizations and Society, 20(2), 93–109. https://doi.org/10.1016/0361-3682(93)E0001-W
- Hyndman, N., & Lapsley, I. (2016). New public management: The story continues. *Financial Accountability & Management*, 32(4), 385–408. https://doi.org/10.1111/ faam.2016.32.issue-4
- Ilmari Rautiainen, A. (2009). The interrelations of decisionmaking rationales around BSC adoptions in Finnish municipalities. International Journal of Productivity and Performance Management, 58(8), 787–802. https://doi.org/10.1108/17410400911000417
- Ingvaldsen, J. A. (2013). Democratic Lean?: Work systems in Norwegian industry. Trondheim: Norges teknisknaturvitenskapelige universitet, Fakultet for samfunnsvitenskap og teknologiledelse, Institutt for industriell økonomi og teknologiledelse.
- Ingvaldsen, J. A., & Benders, J. (2016). Lost in translation? The role of supervisors in lean production. German Journal of Human Resource Management: Zeitschrift für Personalforschung, 30(1), 35–52.
- Ingvaldsen, J., Rolfsen, M., & Finsrud, H. (2012). Lean organisering i norsk arbeidsliv: Slutten på medvirkning? *Magma*, 27(2), 42–50.
- Ingvaldsen, J., Ringen, G., & Rolfsen, M. (2014). Lean på global vandring. In M. Rolfsen (Ed.), Lean blir norsk: Lean i den norske samarbeidsmodellen (pp. 33–47). Bergen: Fagbokforlaget.
- Innovationsrådet. (2012). Lean och systemsyn i stat och kommun-förutsättningar, hinder och möjligheter (Sekretariatsrapport). Stockholm: Author.
- Jung, N., & Kieser, A. (2012). Consultants in the management fashion arena. In M. Kipping & T. Clark (Eds.), The oxford handbook of management consulting (pp. 327–346). New York: Oxford University Press.
- Kieser, A. (1997). Rhetoric and Myth in management fashion. Organization, 4(1), 49–74. https://doi.org/10.1177/135050849741004

Krafcik, J. F. (1988). Triumph of the lean production system. MIT Sloan Management Review, 30(1), 41–52.

Krause-Jensen, J. (2017). Trimming the social body: An analysis of Lean management among family counsellors in a Danish municipality. *Journal of Organizational Ethnography*, 6(2), 68–86.

https://doi.org/10.1108/JOE-01-2017-0005

Lapsley, I., & Knutsson, H. (2016). Modernizing the Public Sector: Scandinavian Perspectives. Milton Park: Taylor & Francis.

Larsson, E. (2012). Managementmoden och popularitetssvängningar–En studie av Lean-konceptet i svensk populärpress 1990–2008. Gothenburg: University of Gothenburg.

Liker, J. K. (1997). Becoming lean: Inside stories of US manufacturers. Boca Raton: CRC Press.

Loenen, V. v., & Schouteten, R. (2016). De implementatie van Lean; Inzicht in inertiefactoren. M & O: Tijdschrift voor Management en Organisatie, 70(2), 20–37.

Madsen, D. Ø. (2014). How do managers encounter fashionable management concepts? A study of balanced scorecard adopters in Scandinavia. International Journal of Management Concepts and Philosophy, 8(4), 249–267. https://doi.org/10.1504/IJMCP.2014.066904

Madsen, D. Ø., Storsveen, M., Klethagen, P., & Stenheim, T. (2016). The diffusion and popularity of Lean in Norway: An exploratory survey. *Cogent Business & Management*, 3(1), 1258132.

Martin, D. (2018). Lean in a cold fiscal climate: The public sector in an age of reduced resources. *Public Money & Management*, 38(1), 29–36.

https://doi.org/10.1080/09540962.2018.1389501 McCann, L., Hassard, J. S., Granter, E., & Hyde, P. J. (2015). Casting the lean spell: The promotion, dilution and erosion of lean management in the NHS. *Human*

Relations. doi: 10.1177/0018726714561697 Mitchell, M., & Jolley, J. (2012). Research design explained. Boston, MA: Cengage Learning.

Modig, N., & Åhlström, P. (2012). This is lean: Resolving the efficiency paradox. Infobel: Rheologica.

Modig, N., & Åhlström, P. (2014). Detta är lean. Infobel: Rheologica Publishing.

Morris, T., & Lancaster, Z. (2006). Translating management ideas. Organization Studies, 27(2), 207–233. https://doi.org/10.1177/0170840605057667

Neumann, S. L., Mothersell, W. M., & Motwani, J. (2014). The need for implementing lean in the public sector. *International Journal of Business Excellence*, 8(1), 104–121.

NOU 2015:1. (2015). Produktivitet—grunnlag for vekst og velferd. Oslo: Departementenes sikkerhets- og serviceorganisasjon, Informasjonsforvaltning.

NOU 2016:3. (2016). Ved et vendepunkt: Fra ressursøkonomi til kunnskapsøkonomi. Oslo: Departementenes sikkerhetsog serviceorganisasjon, Informasjonsforvaltning.

Pettersen, J. (2009). Defining lean production: Some conceptual and practical issues. The TQM Journal, 21(2),

127–142.

https://doi.org/10.1108/17542730910938137

Qu, S. Q., & Dumay, J. (2011). The qualitative research interview. Qualitative Research in Accounting & Management, 8(3), 238–264. https://doi.org/10.1108/11766091111162070

Radnor, Z., & Boaden, R. (2008). Editorial: Lean in public services—Panacea or paradox? *Public Money* &

Management, 28(1), 3–7. Radnor, Z., & Osborne, S. P. (2013). Lean: A failed theory for public services? *Public Management Review*, 15(2), 265– 287. https://doi.org/10.1080/14719037.2012.748820

Radnor, Z., & Walley, P. (2008). Learning to walk before we try to run: Adapting lean for the public sector. Public Money and Management, 28(1), 13–20. Radnor, Z., Walley, P., Stephens, A., & Bucci, G. (2006). Evaluation of the lean approach to business management and its use in the public sector. Edinburgh: Scottish Executive Edinburgh.

Rahbek Gjerdrum Pedersen, E., & Huniche, M. (2011a). Determinants of lean success and failure in the Danish public sector: A negotiated order perspective. International Journal of Public Sector Management, 24(5), 403–420.

https://doi.org/10.1108/09513551111147141

Rahbek Gjerdrum Pedersen, E., & Huniche, M. (2011b). Negotiating lean: The fluidity and solidity of new management technologies in the Danish public sector. International Journal of Productivity and Performance Management, 60(6), 550–566.

https://doi.org/10.1108/17410401111150742

Ringen, G., & Rolfsen, M. (2014). Lean kommune. In M. Rolfsen (Ed.), Lean blir norsk: Lean i den norske samarbeidsmodellen (pp. 118–129). Bergen: Fagbokforlaget.

Rolfsen, M. (2014). Lean blir norsk: Lean i den norske samarbeidsmodellen. Bergen: Fagbokforlaget.

Rolfsen, M., & Ingvaldsen, J. (2012). Unions and the Lean concept. In *ILPC Conference*. Stockholm.

Rossem, A., & Veen, K. (2011). Managers' awareness of fashionable management concepts: An empirical study. *European Management Journal*, 29(3), 206–216. https://doi.org/10.1016/j.emj.2010.10.005

Røvik, K. A. (2007). Trender og translasjoner—ideer som former det 21. århundrets organisasjon. Oslo: Universitetsforlaget.

Røvik, K. A. (2011). From fashion to virus: An alternative theory of organizations' handling of management ideas. Organization Studies, 32(5), 631–653. https://doi.org/10.1137/01708/06611405/626

https://doi.org/10.1177/0170840611405426

Schie, I. (2012). Lean i norske kommuner: En studie om spredning av lean som styringsverktøy, og endringer i organisasjonen som en følge av implementeringen. Kristiansand: Universitetet i Aqder.

- Scorsone, E. A. (2008). New development: What are the challenges in transferring Lean thinking to government? *Public Money and Management*, 28(1), 61–64.
- Shih, T.-H., & Xitao Fan, X. (2008). Comparing response rates from Web and mail surveys: A meta-analysis. Field Methods, 20(3), 249–271.

https://doi.org/10.1177/1525822X08317085

Stentoft Arlbjørn, J., & Vagn Freytag, P. (2013). Evidence of lean: A review of international peer-reviewed journal articles. European Business Review, 25(2), 174–205. https://doi.org/10.1108/09555341311302675

Stentoft Arlbjørn, J., Vagn Freytag, P., & de Haas, H. (2011). Service supply chain management: A survey of lean application in the municipal sector. International Journal of Physical Distribution & Logistics Management, 41(3), 277–295.

https://doi.org/10.1108/09600031111123796

Sturdy, A. (2004). The adoption of management ideas and practices. Theoretical perspectives and possibilities. Management Learning, 35(2), 155–179.

Suarez Barraza, M. F., Smith, T., & Mi Dahlgaard-Park, S. (2009). Lean-kaizen public service: An empirical approach in Spanish local governments. *The TQM Journal*, 21(2), 143– 167.

https://doi.org/10.1108/17542730910938146

Thedvall, R. (2015). Managing preschool the Lean way. Evaluating work processes by numbers and colours. *Social Anthropology*, 23(1), 42–52.

https://doi.org/10.1111/1469-8676.12105

Thedvall, R. (2017). Affective atmospheres of hope: Management model training in public reforms. Journal of Organizational Ethnography, 6(2), 87–99. https://doi.org/10.1108/JOE-01-2017-0003 Wittrock, C. (2015). Reembedding Lean: The Japanese cultural and religious context of a world changing management concept. International Journal of Sociology, 45(2), 95–111. https://doi.org/10.1080/00207659.2015.1061852

Wittrock, C. (2017). Drifting ideas: What diffuses in management fashion studies? (Working paper). Aarhus University.

Womack, J. P., & Jones, D. T. (1996). Lean thinking: Banish waste and create wealth in your organisation. New York, NY: Simon and Shuster. Womack, J. P., & Jones, D. T. (2003). *Lean thinking* (2nd ed.). New York, NY: Free Press.

Womack, J. P., Jones, D. T., & Roos, D. (1990). The machine that changed the world. New York City, NY: Simon and Schuster.

Yadav, O. P., Nepal, B. P., Rahaman, M. M., & Lal, V. (2017). Lean implementation and organizational transformation: A literature review. Engineering Management Journal, 29(1), 2–16. https://doi.org/10.1080/10429247.2016.1263914



© 2017 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license. You are free to:

Share — copy and redistribute the material in any medium or format Adapt — remix, transform, and build upon the material for any purpose, even commercially. The licensor cannot revoke these freedoms as long as you follow the license terms. Under the following terms: Attribution — You must give appropriate credit, provide a link to the license, and indicate if change

 (\mathbf{i})

Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use. No additional restrictions You may not apply legal terms or technological measures that legally restrict others from doing anything the license permits.

Cogent Business & Management (ISSN: 2331-1975) is published by Cogent OA, part of Taylor & Francis Group. Publishing with Cogent OA ensures:

- Immediate, universal access to your article on publication
- High visibility and discoverability via the Cogent OA website as well as Taylor & Francis Online
- Download and citation statistics for your article
- Rapid online publication
- Input from, and dialog with, expert editors and editorial boards
- Retention of full copyright of your article
- Guaranteed legacy preservation of your article
- Discounts and waivers for authors in developing regions

Submit your manuscript to a Cogent OA journal at www.CogentOA.com